Author's response to reviews

Title: Cost of dengue and other febrile illnesses to households in rural Cambodia: A prospective community-based case-control study

Authors:

Rekol Huy (rekolh@cnm.gov.kh)
Ole Wichmann (owichmann@pdvi.org)
Mark Beatty (mbeatty@pdvi.org)
Chantha Ngan (chanthan@cnm.gov.kh)
Socheat Duong (socheatd@cnm.gov.kh)
Harold S Margolis (hsmargolis@pdvi.org)
Sirenda Vong (svong@pasteur-kh.org)

Version: 4 Date: 12 May 2009

Author's response to reviews:

Dear Madam/Sir

We have made the changes regarding each of the suggested revisions except for reference 15:

The URL for reference 15 is correct; however the webpage is temporarily down. I don't know when this is going to be fixed. The home page of the National Bureau of Statistics is http://www.stats.nis.org in which the URL is indicated.

Please let me know what I shall do?

Regards, Sirenda Vong